REVIEW

A systematic review on maternal-to-infant transfer of drugs through breast milk during the treatment of malaria, tuberculosis, and neglected tropical diseases

Francis Williams Ojara ^{1,2}, Aida N. Kawuma¹, Catriona Waitt ^{1,3}*

Infectious Diseases Institute, Makerere University College of Health Sciences, Kampala, Uganda,
Department of Pharmacology and Therapeutics, Gulu University, Gulu, Uganda,
Department of Pharmacology and Therapeutics, University of Liverpool, Liverpool, United Kingdom

* cwaitt@liverpool.ac.uk

Abstract

Background

Exclusive breastfeeding of infants under 6 months of age is recommended by the World Health Organization. In 2021, over 300 million combined incident cases of malaria, tuberculosis, and neglected tropical diseases (NTDs) were reported, predominantly in low-income countries. For many of the drugs used as first-line treatments for these conditions, there is limited knowledge on infant exposure through breastfeeding with poorly understood consequences. This review summarized available knowledge on mother-to-infant transfer of these drugs to inform future lactation pharmacokinetic studies.

Methodology

A list of first-line drugs was generated from the latest WHO treatment guidelines. Using standard online databases, 2 independent reviewers searched for eligible articles reporting lactation pharmacokinetics studies and extracted information on study design, participant characteristics, and the mathematical approach used for parameter estimation. A third reviewer settled any disagreements between the 2 reviewers. All studies were scored against the standardized "ClinPK" checklist for conformity to best practices for reporting clinical pharmacokinetic studies. Simple proportions were used to summarize different study characteristics.

Findings

The most remarkable finding was the scarcity of lactation pharmacokinetic data. Only 15 of the 69 drugs we listed had lactation pharmacokinetics fully characterized. Most studies enrolled few mothers, and only one evaluated infant drug concentrations. Up to 66% of the studies used non-compartmental analysis to estimate pharmacokinetic parameters rather than model-based compartmental analysis. Unlike non-compartmental approaches, model-based compartmental analysis for dynamic characterization of individual plasma



G OPEN ACCESS

Citation: Ojara FW, Kawuma AN, Waitt C (2023) A systematic review on maternal-to-infant transfer of drugs through breast milk during the treatment of malaria, tuberculosis, and neglected tropical diseases. PLoS Negl Trop Dis 17(7): e0011449. https://doi.org/10.1371/journal.pntd.0011449

Editor: Fernando Lopes, Broad Institute Harvard: Broad Institute, UNITED STATES

Published: July 13, 2023

Copyright: © 2023 Ojara et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The authors were supported by a Wellcome Trust Clinical Research Career Development Fellowship awarded to C.W. 222075_Z_20_Z. The funders had no role in the study design, data collection, analysis, or writing of the report.

Competing interests: All other authors declared no competing interests for this work.

and breast milk concentration-time profiles and adequately characterizes variability within and between individuals, using sparsely sampled data. The "ClinPK" checklist inadequately appraised the studies with variability in the number of relevant criteria across different studies.

Conclusions/significance

A consensus is required on best practices for conducting and reporting lactation pharmacokinetic studies, especially in neglected diseases such as malaria, tuberculosis, and NTDs, to optimize treatment of mother–infant pairs.

Author summary

Malaria, tuberculosis, and neglected tropical diseases (NTDs) are prevalent in regions where exclusive breastfeeding is the norm and jointly contributed to more than 300 million incident cases in 2021 alone. Transfer of drugs from the mother to the breastfed infant through breast milk is poorly characterized for most drugs, especially those used for neglected conditions. The clinical consequences of drug exposure such as occurrence of toxicity or predisposition to antimicrobial resistance in the breastfed infants are poorly understood. Infant drug exposure is often estimated from maternal breast milk exposure rather than measurement of infant plasma. In our study, we summarized all existing data on breastmilk transfer of first-line drugs for malaria, tuberculosis, and NTDs. Very importantly, we have highlighted the worrying lack of knowledge and emphasized pertinent study design and data analysis aspects required to improve the accurate conduct and interpretation of lactation pharmacokinetic studies.

Introduction

Malaria, tuberculosis, and neglected tropical diseases (NTDs) are among the leading causes of morbidity and mortality globally. The World Health Organization (WHO) estimated global incidence of malaria in 2020 was 241 million, with 627,000 malaria-related deaths [1]. In the same year, 10 million new tuberculosis infections and 1.5 million tuberculosis-related deaths were estimated [1]. NTDs are a diverse group of 20 conditions mainly prevalent in tropical areas, leading to an estimated 200,000 deaths and the loss of 19 million disability-adjusted life years, respectively, annually [1]. These diseases disproportionately affect poor communities and are significant contributors to the disease burden in sub-Saharan Africa and Southeast Asia.

WHO recommends artemisinin-based combination therapies (ACTs) as first-line for the treatment of uncomplicated malaria (except in pregnant women in their first trimester) [1]. Artemisinin derivatives have short plasma half-lives (approximately 2 hours) and are often coformulated with drugs such as lumefantrine or piperaquine, which have significantly longer half-lives to maintain an antimalarial effect long after the artemisinin derivative has been eliminated [2]. For the treatment of severe malaria, ACTs and quinine are recommended [1].

Tuberculosis is treated using antibiotic combinations extending 6 to 12 months. Rifampicin (R), Isoniazid (H), Pyrazinamide (Z), and Ethambutol (E) are the cornerstone for treatment of drug-sensitive tuberculosis. First-line treatment in new patients with pulmonary tuberculosis comprises 2 months with HRZE followed by 4 months with HR [3]. In rifampicin-susceptible

and isoniazid-resistant tuberculosis, isoniazid is replaced by levofloxacin in a 6-month treatment. Up to 12 additional drugs, including moxifloxacin, bedaquiline, and linezolid are used in different combinations for treatment of multidrug-resistant tuberculosis [3].

NTDs are treated using a diverse range of drugs. Some common antimicrobial agents are used, for example, clarithromycin for the treatment of Buruli ulcers [1] and rifampicin for the treatment of leprosy [1]. For many other NTDs, older and highly toxic drugs are still widely used; for example, melarsoprol, discovered 73 years ago, is still the drug of choice for treating African trypanosomiasis (sleeping sickness) [4]. In some situations, such as for the parasitic infections leishmaniasis and onchocerciasis, treatment and eradication strategies involve the use of mass drug administration campaigns in which an entire population in a geographical location is treated regardless of whether they are infected or not.

WHO recommends exclusive breastfeeding in the first 6 months of life, followed by a combination of breastfeeding and complementary foods [1]. Breast milk is a nutritionally rich matrix containing several micro- and macronutrients and is essential for child nutrition, development, and survival [1,5]. Many mothers take medication while breastfeeding, and where studies have been performed, most drugs found in the maternal blood have been found to be excreted in breast milk to some extent [6,7]. Infant exposure to drugs through breast milk depends on several factors including drug-related physicochemical properties, maternal physiology, and infant feeding patterns (**Fig 1**). A paucity of information on maternal-to-infant transfer of drugs may result in either (i) withholding prescription or choice not to take prescribed medication leading to interruption or undertreatment of maternal conditions; (ii) withholding breastfeeding, which disrupts its benefits for both mother and infant; and (iii) continuing breastfeeding and potentially exposing the infants to small doses of medication [6]. The clinical consequences of an infant's exposure to drugs used to treat malaria, tuberculosis,



Fig 1. Schematic of maternal-to-infant transfer of drugs showing routinely estimated pharmacokinetic exposure factors and different factors that affect infant exposure (Clipart image obtained from https://openclipart.org/search/?query=breastfeeding+mother; date cited: May 30, 2023).

https://doi.org/10.1371/journal.pntd.0011449.g001

and NTDs through breast milk are not clear. Possibilities include adverse reactions, protective exposure, and selection for resistance due to subtherapeutic exposure.

The ratio of drug concentration in maternal breast milk to that in maternal plasma (i.e., the milk-to-plasma [M:P] ratio) provides a crude estimate for infant exposure. The actual infant exposure depends on the M:P ratio, dose and dosing intervals, rate of plasma clearance, and the infant's feeding patterns [8]. The relative infant dose (RID), calculated as a ratio of the daily infant dose to the daily maternal dose, provides a standardized relationship between maternal and infant exposure [9,10]. A theoretical infant milk intake of 150 mL/kg/day is widely used while calculating the RID; however, a dynamic, age-adjusted, infant milk intake model has previously been developed [11]. A RID >10% has been considered the threshold that should raise concern [12] and is widely cited, though this widely accepted gold standard does not account for the inherent toxic potential of the drug and, as a threshold, does not represent an exposure level at which all drugs can be considered toxic. RID calculations may also be based on breast milk volumes and concentrations obtained by completely emptying both breasts over a 24-hour period or by weighing the infant before and after every feed over 24 hours. However, the 2 approaches are not precise and are inconvenient for the mother and infant due to the multiple breast milk collections across 24 hours [13]. M:P ratio and RID are the primary parameters estimated in lactation pharmacokinetic studies. The informativeness of these studies depends on factors such as the number of mother-infant pairs enrolled, the number of samples collected per participant, and the statistical approach for data analysis.

The objectives of this systematic review were to (i) summarize all knowledge available on mother-to-infant transfer of drugs used to treat malaria, tuberculosis (drug-sensitive and rifampicin-resistant/multidrug-resistant, RR/MDR) and NTDs through breastfeeding and (ii) summarize the different mathematical approaches used to characterize maternal-to-infant transfer of drugs used to treat malaria, tuberculosis, and NTDs, to generate evidence on infant exposure and provide recommendations on analysis of lactation pharmacokinetic data for improved characterization of the exposure in a breastfed infant.

Methods

Drugs used to treat malaria, tuberculosis, and NTDs

Using the most recent WHO treatment guidelines for malaria, tuberculosis (including RR/ MDR) and NTDs, we generated a list of drugs used to treat each of the respective diseases (Table 1). For the NTDs, up to 15 diseases were considered based on the WHO's classification [1]. These include Buruli ulcer, Chagas disease, echinococcosis, foodborne trematodiases, human African trypanosomiasis (sleeping sickness), leishmaniasis, leprosy, lymphatic filariasis, mycetoma, chromoblastomycosis and other deep mycoses, onchocerciasis (river blindness), scabies, schistosomiasis, taeniasis/cysticercosis, and yaws and other endemic treponematoses. Other NTDs like rabies, envenoming due to snakebites, Guinea worm infections, dengue, and chikungunya were excluded because they lack specific drugs for their treatment. Once we compiled the list of drugs, we undertook a literature search for pharmacokinetic information on transfer into breast milk.

Literature search strategy

Using the online databases, PubMed, LactMed[®], and SCOPUS, we performed a search using relevant key search terms ("Lactation" or "Breast milk" or "breast feeding" and "pharmacokinetics") for each drug. The above online databases were searched for relevant studies by 2 independent reviewers (F. W. O and A. N. K). Articles were selected based on the inclusion and exclusion criteria detailed below. Articles were explicitly included if the 2 reviewers agreed on

	Recommended treatments
Malaria [1]	Uncomplicated malaria: Artemether, artesunate, lumefantrine, amodiaquine, mefloquine, dihydroartemisinin, piperaquine, sulfadoxine-pyrimethamine, pyronaridine, atovaquone-proguanil, chloroquine, doxycycline Severe (complicated) malaria: Artesunate (IM or IV), quinine <i>P. Vivax</i> or <i>P. Ovale</i> malaria: Primaquine Pregnant women in the first trimester: Quinine, clindamycin
Tuberculosis [3]	Drug-susceptible tuberculosis (patients new on treatment): Rifampicin, isoniazid, pyrazinamide, ethambutol Rifampicin-resistant/multidrug-resistant tuberculosis Kanamycin, capreomycin, levofloxacin, streptomycin, moxifloxacin, bedaquiline, linezolid, clofazimine, delamanid, meropenem, imipenem- cilastatin, amikacin, p-aminosalicyclic acid, clavulanic acid, ethionamide, prothionamide
Buruli ulcers [1]	Rifampicin, clarithromycin, moxifloxacin
Chagas disease [16]	Benznidazole, nifurtimox
Echinococcosis [17]	Albendazole, mebendazole, praziquantel
Trematodiases [16]	Triclabendazole, praziquantel
Human African trypanosomiasis (Sleeping sickness) [4]	<i>T. b. gambiense</i> first stage: Pentamidine, fexinidazole <i>T. b. gambiense</i> second stage: Melarsoprol, erflonithine <i>T. b. rhodesiense</i> first stage: Suramin <i>T. b. rhodesiense</i> second stage: Melarsoprol
Leishmaniasis [<u>18</u>]	Sodium stibogluconate or meglumine antimonate, amphotericin B, miltefosine, paromomycin, pentamidine
Leprosy [1]	Rifampicin, clofazimine, dapsone
Lymphatic filariasis [19]	Ivermectin, albendazole, diethylcarbamazine citrate
Mycetoma [20]	Trimethoprim-sulfamethoxazole, streptomycin, isoniazid, rifampicin, minocycline, amikacin sulfate, amphotericin B, posaconazole, voriconazole, itraconazole
Chromoblastomycosis [21]	5-fluorocytosine, 5-fluorouracil, thiabendazole, amphotericin B, ketoconazole, fluconazole, posaconazole, terbinafine
Onchocerciasis (river blindness) [22]	Ivermectin
Scabies [23]	Ivermectin
Schistosomiasis (Bilhazia) [24]	Praziquantel
	Praziquantel albendazole niclosamide
Laeniasis/cysticercosis [25]	Traziquantei, albenauzote, metosamiae

Table 1. World Health Organization (WHO)-recommended drugs for the treatment of malaria, tuberculosis (including RR/MDR), and neglected tropical diseases (NTDs).

https://doi.org/10.1371/journal.pntd.0011449.t001

their suitability. However, in the case of disagreement, a third reviewer (C.W) was involved to break the tie. No date restrictions were applied. Articles with only abstracts available or those with no official English translation were excluded from the primary compilation. However, non-English articles with an official English translation from alternative sources such as Lactmed [14] were included. For each of the publications, the reviewers extracted the following information: the first author, population studied (healthy volunteers or patients), number of participants (mothers and infants), study design, dose, time postpartum, pharmacokinetic sampling information (plasma or/and breast milk and/or infant plasma, sampling times), mathematical approach for pharmacokinetic analysis, key findings, conclusions, and limitations.

Assessment of study quality

To provide an objective measurement of quality, all selected studies were scored against a 24-item "ClinPK" checklist considered essential for reporting clinical pharmacokinetic studies [15]. This score was developed using Delphi methodology as a "best practice" scale to ensure quality of clinical pharmacokinetic studies; however, not all variables apply to the studies involving breastfeeding mother–infant pairs. Examples of criteria that did not apply to most lactation pharmacokinetic studies were the requirement for detailed reporting of approaches used for extracorporal drug removal in patients on extracorporal drug removal interventions and the criteria requiring reporting of pharmacokinetic parameters such as F (bioavailability), AUC, C_{max}, and T_{max} in studies comparing drug bioavailability between 2 drugs. Additionally, the criteria that required reporting and referencing the formula used for calculating specific pharmacokinetic variables and the criteria that required reporting the specific body weight used in drug dosing were often not applicable in evaluating the lactation pharmacokinetic studies. Each article was assessed on an individual basis with the items considered not relevant excluded from the evaluation of a specific article.

Inclusion and exclusion criteria

We included population lactation pharmacokinetic studies, i.e., studies in which the concentrations of drugs used for the treatment of malaria, tuberculosis, or NTDs were quantified in both maternal plasma and breast milk or in all 3 of maternal plasma, maternal breast milk, and infant plasma. Measuring drug concentrations in both plasma and breast milk enables the characterization of plasma-to-breast milk drug transfer, which enables linking administered drug dose to breast milk exposure. Studies that did not quantify the concentrations of these drugs in both maternal plasma and maternal breast milk or in maternal plasma, maternal breast milk, and infant plasma were excluded. Additionally, case studies were not included in our final list. The systematic review protocol is available at (https://doi.org/10.5281/zenodo. 7974804).

Results

A list of 69 drugs used for the treatment of malaria, tuberculosis (including RR/MDR), and NTDs was generated from the WHO treatment guidelines as summarized in Table 1. Of the 69 drugs listed, only 58% had any information on pharmacokinetics in breast milk available. Of those with information on pharmacokinetics in breast milk, less than half had this information derived from lactation pharmacokinetic studies, i.e., some studies only measured drug concentration in breast milk and not in plasma or took single plasma samples without time recorded relative to dose. In these, it is not possible to characterize the transfer into breast milk, and, therefore, these were not considered lactation pharmacokinetic studies. Fig 2 shows the literature search strategy, and Fig 3 provides a schematic of the workflow adopted in extracting the lactation pharmacokinetic data from evaluated drugs.

Lactation pharmacokinetic studies for drugs used for the treatment of malaria, tuberculosis, and NTDs

Considering the evidence for the use of the 69 target drugs identified from WHO treatment guidelines for tuberculosis (including RR/MDR), malaria, and NTDs, only 18 clinical lactation pharmacokinetic studies, evaluating 15 different drugs, were identified. These studies reported on the following drugs: piperaquine, primaquine, chloroquine, quinine, clindamycin,



Fig 2. Information sources and search strategies.

https://doi.org/10.1371/journal.pntd.0011449.g002

mefloquine, bedaquiline, isoniazid, benznidazole, nifurtimox, albendazole, praziquantel, clofazimine, ivermectin, and azithromycin.

The number of mothers enrolled across the 18 studies was generally low, with a median of 27 mothers (range: 2 to 33). In 4 of the studies, focusing on benznidazole, nifurtimox, primaquine, and bedaquiline, breastfeeding infants were enrolled together with their mothers



Fig 3. Schematic workflow undertaken for extraction of information on lactation pharmacokinetics for drugs used to treat malaria, tuberculosis, and NTDs.

https://doi.org/10.1371/journal.pntd.0011449.g003

[27–30]. In 15 out of 18 studies (83.3%), maternal drug concentrations were measured at multiple time points after drug administration, whereas in the other 3 studies, maternal blood concentrations were measured only at a single time point after drug administration. In the 4 studies that enrolled infants, evaluation of infant blood drug concentrations was performed in 2 studies. In the study by Court and colleagues [30], limited infant plasma sampling was performed, based on 4 infants from the 13 mother–infant pairs. In the study by Gilder and colleagues [29], infant capillary blood concentrations were collected only for evaluation of the hematocrit.

Six of the 18 studies were conducted in women who were not receiving treatment for malaria, tuberculosis, or NTDs but in patients with different disease conditions or even healthy volunteers. Three of the studies that evaluated chloroquine [31,32], ivermectin [33], and praziquantel [34] were undertaken in healthy volunteers, whereas 2 studies evaluated the concentrations of azithromycin in a prophylactic setting in pregnant women about to deliver [35,36]. Eight out of the 18 studies (44.4%) reported the postpartum time points of breast milk sampling, ranging from the time of delivery to as long as 6 months across the different studies.

Across the 18 studies, 66.7% (12 out of 18) used NCA methods for pharmacokinetic evaluation, whereas 6 studies employed model-based compartmental approaches. For NCA, in studies with multiple drug concentrations across time, the most commonly estimated pharmacokinetic parameters for plasma and breast milk drug exposure were T_{max} , C_{max} , and AUC in plasma and breast milk [29,34,37]. Of the 12 studies employing NCA, 5 studies calculated a milk-to-plasma (M:P) ratio based on breast milk and plasma drug concentrations measured at specific time points [27,28,31,37,38]. Four studies calculated the M:P ratio as a ratio of breast milk AUC to the plasma AUC [29,32,34,39], whereas 3 studies did not report calculating a M:P ratio [34,40,41]. M:P ratios calculated from specific time points do not account for differences between plasma and breast milk concentration-time profiles, unlike those generated from AUCs, and might yield misleading findings. For studies that employed model-based compartmental analysis, the M:P ratio was calculated from model-derived breast milk and plasma AUCs [42,43] or by directly modeling the plasma-to-milk drug transfer kinetics [30,44].

Assessment of study quality

Each of the 18 studies were individually assessed for conformity with the best practices for reporting clinical pharmacokinetic studies as recommended by the "ClinPK" guidelines. The 24 items/criteria of the "ClinPK" guidelines were not all relevant across the 18 studies. The criteria requiring that the title of the article identified the drug(s) and the patient population(s) studied was one of the 14 criteria relevant to all 18 studies, whereas the criteria requiring detailed reporting on approaches used for extracorporal drug removal in patients on extracorporal drug removal interventions was among the two not relevant for all 18 studies. The number of relevant criteria was lowest in the studies by Ogunbona and colleagues [31] and Ogbuokiri and colleagues [33], scoring 17 out of 24, and highest in the study by Sutton and colleagues [36] with 22 out of 24. The conformity to the relevant "ClinPK" criteria was highly variable across the 18 studies with the lowest scoring 8 out of 17 in the study by Ogbuokiri and colleagues [33] and the highest score of 20 out of 22 by Sutton and colleagues [36].

A summary of the various study designs, population sampling schedule, and results are reported in <u>Table 2</u> below, and a further description of the findings for each of the chosen drugs is provided. Assessment of the conformity of studies with the "ClinPK" guidelines is provided in the Supporting information (S1 Table).

Table 2. Key fi	ndings in studies evaluating 1	mother-to-iı	nfant transfer of antim	nalarials, antitul	berculosis,	and drugs for neglec	ted tropica	l diseases (NTDs).	
Author	Study design, Population	Number of mothers (infants)	Dose	Time postpartum	Plasma samples (Breast milk)	PK Analysis	ClinPK Score (%)	Conclusion	Limitations
Piperaquine Moore et al [44].	Design: Randomized controlled trial evaluating pharmacokinetics and pharmacodynamics of dihydroartemisinin- piperaquine versus sulfadoxine pyrimethamine- piperaquine combinations for intermittent presumptive treatment Population: Pregnant and non-pregnant women (age- and community matched) with no evidence of severe malaria	27 (0)	320 mg at 0, 24, and 48 hours Standard treatment regimen	Not stated	Yes (Yes)	Compartmental	18/19 (94.7)	M:P ratio = 0.58 AID = 0.41 µg/kg/day RID = 0.004% Breast milk transfer after maternal treatment but exposure for breastfed infants appears safe.	Milk crematocrit was not estimated: Variability in breast milk concentration not evaluated. No infants enrolled
Primaquine Gilder et al [29].	Design: Observational study evaluation the use of standard radical primaquine regimen in preventing relapse of P . <i>vivax</i> malaria infection Population: Women aged ≥ 18 years old with a history of P . <i>vivax</i> history of P . <i>vivax</i> infection and no prior radical cure, together with their breastfeeding healthy (no P . <i>vivax</i> malaria infection) infants	20 (20)	0.5 mg base/kg to nonfasted women once daily for 14 days Standard treatment	Not stated	Yes (Yes)	Non- compartmental	(89.5) (89.5)	M:P ratio = 0.34 (0.12– 0.64) on day 0; 0.37 (0.24– 0.61) on day 13 AID: 2.98 (1.15–9.10) $\mu g/$ kg on day 0; 2.58 (1.06– 8.22) $\mu g/kg$ RID: 0.618% (0.231–1.82) on day 0; 0.517% (0.212– 1.64) on day 13 on day 0; 0.517% (0.212– 1.64) on day 13 Low breast milk transfer, low chances of infant exposure. Treatment exposure the withheld in breastfeeding mothers	Breast milk volume estimated rather than measured across study duration No infantpharmacokinetic data
Chloroquine Ogunbona et al [31].	Design: Observational study evaluating lactation pharmacokinetics and safety in breastfeeding infants when used for prophylaxis Population: Volunteer nursing mothers	11 (0)	Single oral dose of 600 mg base chloroquine Standard dose level for chemoprophylaxis	Not stated	Yes (Yes)	Non- compartmental	9/16 (56.3)	M:P ratio = 5.6 (3.8–9.0) AID = 4.4 ± 2.6 mg RID = 0.70% Breast milk concentrations were greater than plasma concentrations. The average M:P ratio at 24 hours was 6.6 ± 2.4 . The peak concentration was p = 4.4 + 2.6 m/U.	Limited blood/breast milk samples collected No infants enrolled

PLOS NEGLECTED TROPICAL DISEASES

(Continued)

Author	Study design, Population	Number of mothers (infants)	Dose	Time postpartum	Plasma samples (Breast milk)	PK Analysis	ClinPK Score (%)	Conclusion	Limitations
Chloroquine Edstein et al [32].	Design: Observational study evaluating lactation pharmacokinetics when used for prophylaxis Population: Volunteer non-breastfeeding mothers	3 (0)	Single oral dose of 600 mg base chloroquine	2–5 days postpartum	Yes (Yes)	Non- compartmental	12/18	M:P ratio = 1.96, 2.35, and 4.26 RID (%): 2.2, 2.9, 4.2 Chloroquine was excreted in breast milk at concentrations greater than that found in plasma	Limited blood/breast milk samples collected No infants enrolled
Quinine Phillips et al [40]	Design: Observational study evaluating the pharmacokinetics and breast milk excretion of quinine in pregnant and lactating women with falciparum malaria Population: Lactating mothers	30 (0)	10 or 20 mg/kg of quinine sulfate	Late pregnancy/ Early postpartum (Exact time not stated)	Yes (Yes)	Non- compartmental	12/18	In 5 patients treated intravenously, the mean breast milk concentrations was 2.6 mg/L (range: 0.5–3.6) and mean M:P ratio was 0.21 (0.11–0.32) In 25 patients treated orally, the mean concentration was 0.5–8.0 mg/L and 0.31 (0.11–0.53) in 25 women who creceived oral treatment Estimated daily breast milk excretion less than 2-3 mg/day, RID (%) <1%	No infants enrolled in the study
Mefloquine Edstein et al [39]	Design: Observational study evaluating the breast milk excretion of mefloquine lactating women with falciparum malaria Population: Volunteer lactating mothers	2 (0)	250 mg	2–3 days postpartum	Yes (Yes)	Non- compartmental	14/18	The elimination in breast milk than in plasma The M:P ratio values of 0.16–0.13 atter days and 0.27 atter 56 days RID of 3.8%.	Limited number of mothers (only 2) No infants included
Clindamycin Steen and Rane [41]	Design: Observational study evaluating the breast milk excretion of clindamycin in women treated for puerperal anaerobic infections. Population: Lactating mother who have just given birth but not breast feeding their infants	5 (0)	150 mg	One week postpartum	Yes (Yes)	Non- compartmental	9/18	Breast milk concentrations at the end of the dosing interval ranged from <0.5 µg/mL- 3.1 µg/mL. No correlation between milk and plasma concentrations at the end of the dosing interval, however a strong correlated with the area under the plasma	No infant included in the study
									(Continued)

Table 2. (Continued)

Author	Study design, Population	Number of mothers (infants)	Dose	Time postpartum	Plasma samples (Breast milk)	PK Analysis	ClinPK Score (%)	Conclusion	Limitations
Bedaquiline Court et al [30].	Design: Observational study of pharmacokinetics in pregnancy and lactation Population: Pregnant women, aged ≥18 years treated for RR-TB, together with their breastfeeding infants	13 (13)	200 mg dosed 3 times a week, affer the 2-week loading dose. Standard treatment	13 women (≥28 weeks) and 6 women (6 weeks postpartum)	Yes) (Yes)	Compartmental analysis	(88.2) (88.2)	M:P ratio = 13.6 (% RSE = 10.1) for bedaquiline and 4.84 (RSE = 5.10) for the metabolite AID = 0.816 mg/kg/day for bedaquiline and 0.07/ mg/kg/day for the metabolite RID = 66.9% The drug significantly accumulates in breast milk; breastfed infants receive mg/kg doses equivalent to maternal doses.	Unbound drug concentrations or albumin levels were not measured, possibly leading to low plasma concentrations. High rate of participant loss to follow up limiting sample size. Limited infant pharmacokinetic data, based on only 4 infants.
Isoniazid Singh et al [42].	Design: Observation study of lactation pharmacokinetics for mothers on standard anti tuberculosis treatment Population: Lactating women with tuberculosis, aged 18–28 years	7 (0)	300 mg single dose daily Standard treatment	Not stated	Yes (Yes)	Compartmental	15/19 (78.9)	M:P ratio = 0.89 (95% CI: 0.70-1.1) AID = 89.9 µ/kg/day (95% CI: 65.6-114) RID = 1.2% (95% CI: 0.9, 1.5) The data suggest that isoniazid therapy is safe during breastfeeding	Few mothers included in the study No infants enrolled
Benznidazole Garcia- Boumissen et al [27].	Design: Observational study of lactation pharmacokinetics and the risk of infant exposure Population: Lactating women with chronic Chagas disease aged 20–34 years	10 (10)	5–8 mg/kg/day twice daily for 30 days Standard treatment	Not stated	Yes (Yes)	Non- compartmental	13/17 (76.5)	M:P ratio = 0.52 (0.3- 2.79) AID = 0.65 mg/kg/day RID = 12.3% (5.5-17.0) Limited drug exposure in breast milk hence may be compatible with breast feeding	No infant pharmacokinetic data Small number of infants were enrolled hence not possible to accurately evaluate adverse effects in infants
Nifurtimox Moroni et al [28].	Design: Observational study of lactation pharmacokinetics and safety Population: Lactating women with chronic Chagas disease age 17–36 years	10 (10)	8–12 mg/kg/day Standard treatment	Not stated	Y es (Yes)	Non- compartmental	12//18	M:P ratio = 16 (8.75-30.3) AID = 0.50 mg/kg/day (1QR: 0.20-0.69) RID = 6.70 (1QR: 2.35- 7.19) Nifurtimox may be compatible with breastfeeding due to limited drug transfer into breast milk, and low overall infant exposure.	No infant pharmacokinetic data Few infants enrolled, hence not possible to accurate evaluate adverse effects

Author	Study design, Population	Number of mothers (infants)	Dose	Time postpartum	Plasma samples (Breast milk)	PK Analysis	ClinPK Score (%)	Conclusion	Limitations
Albendazole Abdel-tawab et al [37].	Design: Observational study of lactation pharmacokinetics and infant exposure Population: Breastfeeding women aged 18–40 years in a lymphatic filariasis control program	33 (0)	400 mg Standard treatment	2 weeks to 6 months	Yes (Yes)	Non- compartmental	13/18 (72.2)	M:P ratio = 0.9 (0.2–6.5) for albendazole and 0.6 (0.1–1.5) for albendazole sulphoxide Albendazole (and the sulphoxide) attains low concentrations in breast milk unlikely to harm the breastfed infant	Limited (only one) maternal plasma concentration measured. No infants enrolled
Praziquantel Bustinduy et al [43].	Design: Randomized controlled trial for evaluating pharmacokinetics and safety of praziquantel in pregnancy Population: Lactating women infected with Schistosomiasis Japonicum	15 (0)	30 mg/kg twice over 4 hours Standard treatment with same overall dose	Not stated	Yes (Yes) 0.028	Compartmental analysis	(90)	M:P ratio = NE AID = 0.028 mg/kg/day RID = 0.028 mg/kg/day Women in early pregnancy had increased clearance and lower exposure. Significantly lower breast milk exposure than the dose required for antischistosomal activity. Breastfeeding should not be stopped or delayed	No infants enrolled
Praziquantel Putter and Held [34]	Design: Randomized trial evaluating lactation pharmacokinetics of praziquantel at various dosing schedules Population: Healthy lactating women	10 (0)	50 mg/kg once (5 women); 20 mg/kg 3 times at 4-hour intervals (5 women) Standard treatment dose	Not stated	Yes (Yes)	Non- compartmental analysis (Exact method not stated)	(62.5)	The mean plasma exposure was 3.8 times the mean breast milk exposure. AID = 27.9 µg/day RID = 27.9 µg/day RID = 0.0007% Milk does not represent a deep distribution compartment but readily equilibration obviously takes place by passive diffusion and not active secretion	No infants enrolled
Clofazimine Venkatesan et al [38].	Design: Observational study of lactation pharmacokinetics and infant exposure to clofazimine as part of multiple drug treatment Population: Female leprosy patients, aged 19–35 years	8 (0)	50 mg daily (5 women); 100 mg on alternate days (2 women); 100 mg daily (1 woman) Standard treatment	Not stated	Yes (Yes)	Non- compartmental	16/17 (94.1)	M:P ratio = 1.4 \pm 0.08 at 4-6 hours after last daily dose AID = 0.199 \pm 0.013 mg/ kg/day RID = 22.11 \pm 1.90% In spite significant breast milk exposure (infant ingestion of 0.225 mg/kg/ day), discontinuation is not recommended in breastfeeding mothers till further toxicity studies	Plasma data only available at a single time point. No infants enrolled
									(Continued)

Table 2. (Continued)

Table 2. (Cont	inued)								
Author	Study design, Population	Number of mothers (infants)	Dose	Time postpartum	Plasma samples (Breast milk)	PK Analysis	ClinPK Score (%)	Conclusion	Limitations
Ivermectin Ogbuokiri et al [33].	Design: Controlled trial of lactation pharmacokinetics and infant exposure Population: Healthy mothers whose babies died at birth	4 (0)	150 µg/kg Standard treatment for river blindness	Not stated	Yes (Yes)	Non- compartmental	7/16 (43.8)	AID = 2.75 µg/kg RID = 2.75% Given the additional benefits of ivermectin and its tolerably, low levels in breast milk, exclusion of lactating mothers should be discontinued.	No infants included in the study. Mothers were not breastfeeding—did not sample mature milk
Azithromycin Sutton et al [36].	Design: randomized controlled trial evaluating perinatal pharmacokinetics of azithromycin after a single preincision administration Population: Women undergoing planned cesarean delivery at 37 weeks of gestation (or more)	30 (0)	500 mg	At delivery	Yes (Yes)	Compartmental (2-compartment model)	(90.5)	AID = 340 µg/day RID = 0.07% High and sustained breast milk concentrations obtained after a single dose. Accumulation of the drug in breast milk from simulation-based studies	No infants enrolled
Azithromycin Salman et al [35].	Design: randomized placebo -controlled Phase III trial of a single dose of azithromycin versus placebo for lactation pharmacokinetics and infant safety Population: Women in labor	20 (0)	2 8	Within 1 month postpartum	No (Y es)	Compartmental	16/18 (88.8)	AID = 0.7 mg/kg/day RID = 2.5% Breast milk exposure may exceed the nominal 10% safety limit in a large proportion of neonates even after a single dose	M:P ratio used was not derived from women in this study. More complex (mechanistic models) were not evaluable due to the limited data. No infants enrolled
AID abcoluted	oily infant dose. M.D. mills to	nlaema ratio	DID valativa infant d	0.00					

AID, absolute daily infant dose; M:P, milk-to-plasma ratio; RID, relative infant dose.

The majority of these studies did not measure concentrations of the drugs in infant plasma limiting the characterization of the actual level of exposure in infants.

https://doi.org/10.1371/journal.pntd.0011449.t002

Piperaquine

Twenty-seven women received 3 doses containing 320 mg of piperaquine (combined with dihydroartemisinin or sulfadoxine-pyrimethamine) administered at 0, 24, and 48 hours after taking a detailed medical history and physical examination of the patient. Breast milk was sampled on days 1, 2, 3 to 5, 7 to 11, and 14 to 17 after delivery. Changes in plasma and breast milk concentrations were characterized using an integrated breast milk-plasma model, with plasma concentrations described as linearly changing across time, whereas the breast milk concentrations were estimated from plasma concentrations using the M:P ratio. The estimated absolute and relative cumulative infant dose was 22 μ g and 0.07%, respectively, corresponding to an absolute and RID of 0.41 μ g/kg/day and 0.004%, respectively [44].

Primaquine

The levels of primaquine in maternal blood and breast milk and blood exposure of a breastfed infant were evaluated in 20 mother–infant pairs after administering 0.5 mg base/kg of the drug daily for 14 days to non-fasted women [29]. From samples collected on days 0, 3, 7, and 13, the pharmacokinetics of the primaquine and its metabolite (carboxyprimaquine) were evaluated. A delayed distribution into breast milk was observed, with concentrations peaking 1 hour later in breast milk compared to plasma. The infant was estimated to ingest up to 0.042 mg/kg over the 14 days of treatment, corresponding to 2.98 μ q/kg/day (only 0.6% of hypothetical infant daily dose of 0.5 mg/kg). The breast milk pharmacokinetics were similar on days 0 and 13, whereas those of carboxyprimaquine varied between the 2 days. The infant breast milk exposure was negligible with only a single sample, collected in one of the infants on day 7, being above the lower limit of quantification. In addition to the negligible infant exposure, no adverse events were reported in infants [29].

Chloroquine

Two studies [31,32] evaluated the breast milk excretion of chloroquine and its major metabolite, desethylchloroquine, after a single oral dose of 600 mg. In the first study [31], 11 female lactating volunteers were evaluated. From plasma and breast milk samples collected at 0, 3, and 24 hours postdose, concentrations of chloroquine in breast milk were always greater than that in plasma, with an average M:P ratio of 6.6 ± 2.4 and 1.5 ± 0.6 for desethylchloroquine. The estimated maximum RID was 0.7%. From urine sampled in 4 of the neonates between 12 and 24 hours, chloroquine concentrations of 3.97 ± 1.6 were observed, whereas those of desethylchloroquine were 0.44 ± 0.32 [31]. In the second [32] study, 3 women were administered chloroquine 2 to 5 days postpartum, and 6 blood and breast milk samples per patient were obtained over a period of 10 days from the time of drug administration. The M:P ratio values of 1.96, 2.35, and 4.26 were calculated for chloroquine, corresponding with percentage doses excreted in breast milk of 2.2, 2.9 and 4.2. The M:P ratio values for desethylchloroquine were 0.54, 0.80, and 3.89.

Quinine

The breast milk excretion of quinine was studied in 30 lactating mothers treated with an initial parenteral dose of 10 or 20 mg/kg of quinine sulfate [40]. The period of start of lactation spanned from within the previous 24 hours to up to 10 days ago. Five patients received between 2 and 7 doses intravenously and had mean breast milk concentrations of 2.6 mg/L (range: 0.5 to 3.6) and mean M:P ratio of 0.21 (0.11 to 0.32). Twenty-five patients who received

oral treatment for 1 to 10 days had mean breast milk concentration of 0.5 to 8.0 mg/L and mean M:P ratios of 0.31 (0.11 to 0.53).

Clindamycin

One study [41] evaluated the breast milk excretion of clindamycin. Five mothers received clindamycin 3 times a day starting immediately after childbirth. The lactation pharmacokinetics was studied within a single dosing interval after a week of treatment. The breast milk concentrations at the end of the dosing interval ranged from <0.5 μ g/mL to 3.1 μ g/mL. No correlation between milk and plasma concentrations was observed at the end of the dosing interval; however, the milk concentrations strongly correlated with the area under the plasma concentration-time curve. As the volumes of milk collected at the end of the dosing intervals were not reported, it is not possible to compute the amounts of drug excreted and the RID.

Mefloquine

One study evaluated the breast milk excretion of mefloquine [39]. Two women who received 250 mg of mefloquine 2 to 3 days postpartum were studied. Blood was sampled predose, at 4 hours, and 1, 2, 4, 14, 28, 42, and 56 days postdose, whereas breast milk was sampled at 0 hours, 4 hours, and 1, 2, and 4 days postdose for both women. Additionally, at 14, 28, 42, and 56 days postdose for one of the women. The breast milk elimination of mefloquine was slower than that in plasma. The M:P ratio was 0.16 and 0.13 in either of the 2 women in the first 4 days and 0.27 after 56 days. It was estimated that a 4-kg infant with a daily milk intake of 1 L would consume 0.08 mg/day, and 0.56 mg/week, corresponding to a RID of 3.8%.

Bedaquiline

A longitudinal pharmacokinetic study of this drug was conducted in 13 pregnant women aged 30 years (interquartile range (IQR): 25 to 37) [30]. During their third trimester of pregnancy, they were dosed with 200 mg of bedaquiline 3 times a week as part of a standard regimen for treating rifampicin-resistant tuberculosis. Blood was sampled predose and at 2, 4, and 6 hours postdose in their third trimester of pregnancy (\geq 28 weeks) and at 6 weeks postpartum, whereas breast milk was sampled at 6 weeks postpartum. Using a population pharmacokinetic model developed to describe this data, the M:P ratio was estimated as 13.6 (% relative standard error = 10.1) for bedaquiline and 4.84 (% relative standard error = 5.10) for M2 (a metabolite of bedaquiline). The estimated infant doses were 0.816 mg/kg/day for bedaquiline and 0.07 mg/kg/day for M2. The estimated maternal daily dose of 1.22 mg/kg/day would translate into a RID of 66.9%. The bedaquiline and M2 concentrations in the breastfed infants were similar to that in maternal plasma, i.e., less than 0.02 mg/mL for both bedaquiline and the metabolite [30]. This extensively high infant exposure (RID >10%) emphasizes the need for lactation pharmacokinetic studies.

Isoniazid

Singh and colleagues [42] evaluated steady-state pharmacokinetic exposure in 7 women who received 300 mg of isoniazid once daily for at least 34 days as part of standard treatment for tuberculosis, also including rifampin and ethambutol. From breast milk sampled between 0 to 4 hours after a dose, the maximum concentrations were 2 to 6.7 mg/L, occurring 1 hour after the dose and the estimated exposure of a breastfed infant was 89.9 μ g/kg/day corresponding to a RID of 1.2% [42]. Two earlier studies also reported the distribution of isoniazid in breast

milk. In the first study, 30 lactating mothers each received a single oral dose of 200 mg of isoniazid. The average peak breast milk concentration from samples obtained between 2 and 7 hours after dose was 2.1 mg/L at 2 hours in 6 women and 0.46 mg/L at 7 hours in 3 women. In the second study, 3 mothers who received single oral doses of 300 mg and 600 mg of isoniazid reported similar time to peak drug concentrations (3 hours after dose) but higher maximum concentrations with the 600 mg compared to the 300 mg dose, i.e., a range of 9 to 10.6 mg/L versus 5.4 to 5.5 mg/L [14].

Benznidazole

Ten lactating mothers treated with 5 to 10 mg/kg/day of benznidazole twice daily for 30 days were evaluated. From the 16 breast milk samples collected from 10 patients, the median breast milk benznidazole concentration was 3.8 mg/L. The estimated median calculated infant dose was 0.65 mg/kg daily corresponding to about 10% of the dose used to treat infants with Chagas disease and yielding a 12.3% RID [27]. Additionally, the breast milk distribution of benznidazole was also evaluated in a study developing a bioanalytical method for breast milk quantification [45]. From 8 breast milk samples from different lactating mothers treated with 5 to 10 mg/kg/day of benznidazole every 12 hours for 30 days, concentrations ranging from non-quantifiable (<0.88 mg/L) to 7.1 mg/L were observed between the fourth and 10th day. Limited clinical extrapolations can be made from this study due to the few samples collected and the lack of information on dosing times [45].

Nifurtimox

In a single study, 10 breastfeeding mothers with chronic Chagas disease received 8 to 12 mg/ kg/day of nifurtimox 3 times daily for 30 days. From 17 steady-state (days 4 to 21) breast milk concentrations obtained across all the mothers, a median concentration of 0.30 mg/L was reported. The estimated median infant daily dose was 0.5 mg/kg/day (IQR: 0.20 to 0.69) representing a median RID of 6.70%. No growth-, behavioral-, or weight-related adverse events that could be associated with nifurtimox were reported in all the infants enrolled in this study [28].

Albendazole

The distribution of albendazole and its metabolites (sulphoxide and sulphone) in breast milk was evaluated in 20 lactating women after a single oral dose of 400 mg in a national lymphatic filariasis control program [37]. From serial breast milk samples collected from 20 patients at predose, 6, 12, 24, and 36 hours postdose, the maximum calculated concentration of albendazole sulphoxide was 352 ng/mL observed at 6.9 hours postdose. Using the average of reported breast milk concentrations, an infant milk intake of 0.15 L/kg/day would result in a dose of 0.00218 mg/kg/day and a RID of less than 0.05%. The AUC between 0 and 24 hours, and 0 and 36 hours were 3,932.2 ng•h/mL and 5,190.3 ng•h/mL, respectively [37].

Praziquantel

In a study by Putter and Held [34], breast milk exposure of praziquantel was evaluated in 10 healthy lactating women. The amount of drug excreted in breast milk was estimated as a product of breast milk volume and associated concentrations within a specific sampling interval. In 5 women treated with a single 50 mg/kg dose, the maximum amount excreted was 8.5 μ g, 2 hours postdose, and the mean of the total amount of drug excreted at 24 hours across all individuals was 27.4 μ g (RID = 0.00087%). In the other 5 women treated with 3 doses, 20 mg/kg every 4 hours, maximum amounts of drug were excreted at 4 and 10 hours after the first dose,

and the mean of the total amount of drug excreted at 24 hours across all individuals was $27.4 \ \mu g \ (RID = 0.00087\%)$.

In another study by Bustinduy and colleagues [43], praziquantel PK was evaluated in 15 lactating women infected with *Schistosoma japonicum* after two 30 mg/kg doses of praziquantel given 3 hours apart. From serial breast milk samples obtained at 3, 6, 9, 12, 15, and 24 hours postdose, an integrated population pharmacokinetic model describing reversible transfer of the drug between plasma and breast milk was developed. The average concentration in breast milk over 24 hours was 0.185 mg/L, and the estimated daily infant intake was 0.028 mg/Kg, translating into a RID of 0.05%.

Clofazimine

A study by Venkatesan and colleagues [38] reported the breast milk distribution of clofazimine in 8 female leprosy patients in their early lactation period (\leq 4 months) who were evaluated after treatment with doses between 50 mg and 100 mg daily (1 patient received 100 mg daily; 2 patients received 100 mg on alternate days; 6 patients received 50 mg daily) for periods between 1 and 18 months. From breast milk sampled between 4 and 6 hours postdose, clofazimine concentrations ranged from 0.8 to 1.7 µg/mL. The estimated infant daily dose ranged between 0.120 and 0.255 mg/kg/day representing an RID of 13.5% to 30% of the last maternal dose [38].

Ivermectin

In the study by Ogbuokiri and colleagues [33], the distribution of ivermectin in breast milk was evaluated in healthy lactating mothers after a single dose of 150 μ g/kg of ivermectin. Breast milk drug concentrations between 1 and 72 hours postdose was 9.85 ng/mL (range: 4.2 to 20.6). The maximum breast milk concentration was 14.13 ng/mL at 4 hours for 2 women, 6 and 12 hours each of the other two. The estimated daily infant breast milk consumption of ivermectin was 2.75 μ g/kg, translating into a RID of 2.75%, with a limited risk of negative consequences for the infant.

Azithromycin

Two studies [35,36] both employed model-based approaches to characterize the breast milk distribution of azithromycin administered in women just before labor. The estimated absolute and relative daily infant doses after a single maternal oral dose of 2 g were 0.7 mg/kg/day and 2.5% [35]. In women who received 500 mg within an hour before cesarean delivery, a sustained breast milk concentration was observed up to 48 hours, with a median of 1,713 ng/mL reported 30.7 hours postdose [36]. Model-based simulations further suggested that with doses (500 mg each) given once every 12 hours, one would yield steady-state concentrations after about 3 days and an exclusively breastfed infant would receive about 340 µg/day, translating into a RID of 0.07%.

Discussion

The most remarkable finding of this review is that lactation pharmacokinetic studies have not been undertaken for the majority of drugs, even though these drugs are widely used in breastfeeding women. Of the 69 drugs identified as widely used to treat malaria, tuberculosis, and NTDs, only 15 drugs across 18 studies fit our inclusion criteria. Approximately half (58%) of the 69 drugs had some information regarding breast milk concentrations reported in literature, but in many situations, this information was not derived from lactation pharmacokinetic studies. Instead, information was drawn from other research activities including case reports or analyses done for bioanalytical assay development. This reluctance to conduct lactation pharmacokinetic studies is not new. A review by Larsen and colleagues [46] found that of 192 drugs that had some information on breast milk concentrations, only 69 drugs (from 78 studies) had sufficient data to calculate a M:P ratio. They mention that for most of the studies reporting breast milk concentrations, only a single sample had been collected, precluding the calculation of breast milk AUC (AUC_{breast milk}) and the M:P ratio, which is calculated as AUC_{breast milk} / AUC_{plasma}. The M:P ratio is more reliable when computed from exposure across the entire dosing interval (i.e., AUC). Considering that breast milk and plasma drug concentrations are often not parallel to each other, computing the M:P ratio based on a single time point can lead to misleading or erroneous conclusions about the distribution of the drug in breast milk [9]. In fact, a graphical comparison of plasma and breast milk data may help elucidate the dynamics of drug transfer and generate hypotheses on the underlying factors that influence maternal-to-infant transfer of drugs.

A second observation is that few mothers were enrolled in the selected studies, with a median number of 27 (range: 2 to 33). The low numbers constrain the accurate characterization of interindividual variability in the drug's pharmacokinetics and make it difficult to identify patient factors that influence the pharmacokinetics of the drugs. Only one of the 18 studies evaluated infant plasma drug concentrations in addition to maternal blood and breast milk drug concentrations. Consequently, in 17 studies, maternal-to-infant transfer of drug (infant exposure) was estimated from the plasma and breast milk drug concentrations. While this approach offers an informed estimate, the more robust and conclusive way to ascertain infant exposure would be to evaluate concentrations in the breastfed infant. This lack of drug concentrations in infants limits the ability to directly correlate maternal plasma and breast milk concentrations with infant exposure. Caution should be made while extrapolating infant exposure from breast milk exposure. Firstly, the bioavailability from maternal breast milk can only be considered if the pharmacologically active species in breast milk is the parent compound that requires no biotransformation for activation. Secondly, the immaturity of enzymatic elimination pathways can drastically alter exposure in infants compared to adults, for example, the half-life of dolutegravir, a UGT1A1 substrate, was estimated to be approximately 4-fold longer in neonates compared to adults [47]. Thirdly, genotypic differences in drug-metabolizing enzymes between the mother and the infant may lead to differences in drug exposure between the mother and the infant. Additionally, several additional unknown factors may lead to variability in drug profiles between different infants. Nevertheless, infant exposure can still be estimated based on breast milk data alone. For example, Rodari and colleagues [48] estimated infant exposure to ivermectin based on 6 serial breast milk samples collected predose to 24 hours postdose. Quantifying maternal plasma drug concentrations provides a link between the administered dose and the ultimate process of excretion into breast milk and characterizing patient- or treatment-related factors that affect plasma concentrations will also highlight factors that affect breast milk excretion of the drug.

A third observation relates to the data analysis approaches used. Non-compartmental analysis (NCA) was used in 12 out of the 18 studies to characterize plasma and breast milk drug pharmacokinetics. Compared to compartmental analysis, NCA offers a faster means of estimating pharmacokinetic parameters such as C_{max} , T_{max} , elimination half-life, and the AUC [49]. However, a major drawback to its use is that it requires intensive pharmacokinetic sampling in each individual to describe the complete concentration-time profile and efficiently estimate the pharmacokinetic parameters [49]. NCA makes no assumptions about the disposition of a drug, which compartmental approaches often do. The use of NCA usually precludes the investigation of covariates (individual factors) associated with variability in the pharmacokinetics of a drug and thus its transfer from plasma to breast milk. Conversely, the

compartmental analysis approach does not require as intensive sampling as NCA in each individual. However, compared to NCA, it is more computationally intensive, technically more complex, and, hence, requires a greater amount of training and depends on specific underlying assumptions on the distribution of the drug in the body to characterize the drug's pharmacokinetics.

Nonlinear mixed-effects (NLME) modeling is the commonest compartmental analysis framework and was used to characterize the pharmacokinetics of isoniazid, azithromycin, and piperaquine [36,42,44]. NLME models simultaneously characterize typical profiles of change in drug concentrations across time within a population and the different levels of variability (within a specific patient and across different patients in a population) [50]. A key strength of NLME modeling is that it allows for robust and precise estimation of pharmacokinetic parameters even with sparse data (e.g., 3 samples per patient within a dosing interval for a study population consisting of approximately 50 patients). This reduces the burden of frequent blood draws and increases the feasibility of studies in populations where these would be difficult, for example, in pediatric or severely ill populations. Overall, compared to NCA, it reduces the logistical burden in executing the study. Additionally, these models can take various functional forms, allowing integration of knowledge on relevant physiological processes or observed trends in the data to better characterize concentration-time profiles. Fig 4 compares non-compartmental analyses showing the level of informativeness and the associated pros and cons.

PBPK models integrate prior diverse knowledge for a more dynamic mechanistic characterization of milk-to-plasma transfer of the drug [51,52]. To predict infant plasma drug concentrations, different infant maturation processes can be integrated in PBPK models to improve predictions.



Fig 4. Schematic depiction of the plasma and breast milk changes in drug concentrations across time comparing non-compartmental and compartmental analysis. Left panel: Non-compartmental analysis; right panel: Compartmental analysis; open circles: Individual drug concentration data, plasma (black) and breast milk (blue); Lines: Individual drug concentration-time profiles in plasma (black) and breast milk (blue); solid lines: Population drug concentration-time profile in plasma (black) and breast milk (blue).

https://doi.org/10.1371/journal.pntd.0011449.g004

The fourth observation was that 7 of the 18 studies were not from malaria, tuberculosis, or NTDs therapeutic areas but included patients with different disease conditions or even healthy volunteers. For example, a single 30 mg/kg (up to 2.0 g) dose of azithromycin is administered in the treatment of yaws, whereas, in this review, azithromycin was studied at 500 mg in women due for planned cesarean delivery at term [36] or at 2.0 g in women in labor [35]. Considering that drug doses and dosing schedules often differ across therapeutic areas, conclusions drawn from a pharmacokinetic study implemented in one therapeutic area may not translate to another area. Differences in doses and dosing schedules may lead to differences in pharmacokinetic exposure. This may be due to processes such as enzyme saturation, which can lead to markedly increased exposure with increasing doses or induction of enzymes leading to reduced exposure over time (single dose versus repeat dosing). Additionally, disease-related physiological changes might affect a drug's pharmacokinetics, i.e., drug-disease interactions, which cannot be accounted for if the drugs are studied in different therapeutic areas. Furthermore, great care needs to be taken when translating results derived from studies in pregnant women to those in non-pregnant women. Pregnancy is known to cause many physiological changes that affect drug disposition including inducing drug metabolizing enzymes and drug transporters, altering amounts of proteins like albumin (which bind drugs), and increasing the volume of distribution of some drugs. As such, plasma-to-milk transfer of drugs in a pregnant or very early postpartum women might be different from that in a woman who is >6 weeks postpartum. On the other hand, if a drug (such as azithromycin in this case) is most likely to be used in pregnant women during or shortly before delivery, then it makes sense to determine the M:P ratio in these women to ascertain exposure in the neonate, more so if the drug has a long half-life and might be used in days/weeks leading up to delivery.

A key limitation that we observed across the studies is that many neglected to collect/report information on the postpartum times of pharmacokinetic sampling. Breast milk is a nutritionally rich matrix with several nutrients whose compositions vary with the time postpartum. It has been reported that the protein and fat concentrations in breast milk show a positive correlation with lactation from the first to the 48th month, whereas the carbohydrate concentrations show a negative correlation with lactation in the same time frame [5]. As such, fat-soluble drugs may show a higher distribution in hind milk (high fat content) compared to fore milk (low fat content) [9] since they may dissolve in the lipid droplets as they form in the alveolar epithelial cells leading to cosecretion of the drugs in breast milk.

The "ClinPK" checklist [15] was used to assess conformity of the selected studies with the minimum standards for reporting clinical pharmacology studies. Standardized guidelines facilitate transparent and complete reporting of clinical studies, thereby providing guidance to researchers conducting such studies and increasing the usability of their results by clinicians and researchers [15,53]. Clinical pharmacology trials are very diverse, sometimes not randomized, with quite different objectives, designs, and analysis methods [53]. The different guidelines for reporting clinical studies attempt to account for the diversities in study characteristics. The CONsolidated Standards Of Reporting Trials [54] is currently the most widely used criteria for reporting randomized control trials. For clinical pharmacokinetic studies that employ population-based analyses, different guidelines [55,56] have been proposed, whereas for clinical pharmacology studies, the 24-item "ClinkPK" checklist has been proposed [15]. Though there is a huge overlap in the criteria specified across the different reporting guidelines [15,54–56], divergencies result from differences in study objectives, patient inclusion, and statistical methodology. Not all the 24 items of the "ClinPK" checklist were relevant to the 18 studies. The criteria requiring detailed reporting of approaches used for extracorporal drug removal in patients on extracorporal drug removal interventions was not relevant in all 18 studies. The criteria that required a description of covariates included in a population

pharmacokinetic model was only relevant in 5 out of the 18 studies. As such, each study was evaluated individually using only the relevant criteria. A lower percentage "ClinPK" score does not mean low informativeness of a study: For any 2 studies, the lack of conformity to the same number of criteria leads to a lower percentage "ClinkPK" score in the study with fewer relevant criteria.

Two recent guidelines for lactation studies have been published [57,58]. The USFDA in 2019 outlined a draft guideline for industry for conducting clinical lactation studies [57]. The guideline outlines key considerations for researchers, including ethical considerations, participant enrollment (mothers and infants), sample collection for both plasma and breast milk, data analysis, and estimation of infant exposure, and resultant adverse effects. A more recent guideline by Anderson suggested 8 major elements for reporting medication use during lactation [58]. The guideline covers pharmacokinetics and pharmacodynamics of lactation studies and seeks to identify a causal link between the pharmacokinetics of a drug and the observed effect on the infant. Pharmacodynamics elements account for 50% of the major elements in this guideline, all of which are not relevant for pharmacokinetic studies. The two guidelines can provide the initial template for further development. Criteria such as calculation of M:P ratio based on breast milk and plasma AUCs, reporting postpartum time of pharmacokinetic sampling, and infant plasma sampling would be included in a "*lactation pharmacokinetic score.*"

More clinical lactation studies have been conducted for antiretroviral drugs than any other infectious disease. In a review, Waitt and colleagues [59] reported up to 24 lactation studies covering different classes of first-line antiretroviral drugs. Of all drug classes, nucleoside reverse transcriptase inhibitors showed the highest breast milk accumulation, followed by non-nucleoside reverse transcriptase inhibitors and protease inhibitors [59]. A wide intra- and interstudy variability due to differences in pharmacokinetic sampling times, drug concentration assays, statistical methods used for data analysis, and biological differences between populations were reported, all that affect the interpretation of observed findings [59]. A more recent study characterized the breast milk transfer of the relatively newer antiretroviral drugs including dolutegravir, raltegravir, bictegravir, rilpivirine, and darunavir, showing high rilpivirine transfer, moderate to high transfer of raltegravir, and low transfer of bictegravir and dolutegravir [60]. Bictegravir and dolutegravir showed high infant concentrations despite the low breast milk concentrations observed [60], which, in the case of dolutegravir, relates to the high transplacental transfer of drug (cord: maternal blood ration of 1.2) and prolonged infant clearance due to immaturity of UGT1A1 [61].

The LactMed[®] database provides a useful summary of clinically relevant information from lactation studies for several drugs used for treating malaria, tuberculosis, and NTDs. Lactmed summarizes data on maternal breast milk transfer and infant exposure and attempts to link this exposure with treatment outcomes on the infant, even when the data informing this overall judgment are few and sparse. This systematic review provides a broader view of different aspects of lactation studies, also including study design aspects and a summary of overall quality of the study, patient characteristics, as well as maternal plasma exposure, breast milk transfer, and infant exposure, hence providing a platform for optimizing lactation studies.

The eminent risk of unnecessary infant exposure to study drugs and possible harm from sampling blood in breastfed infants are often cited as ethical obstacles to pursuing lactation pharmacokinetic studies involving both the mother and the infants and a possible reason for not sampling from infants in most of the studies in this review. With exception of the study by Ogbuokiri and colleagues [33] that enrolled consented mothers who had lost their babies at birth, hence no risk of infant exposure, the other healthy volunteer studies [33–35, 37] enrolled breastfeeding mothers, without mention of cessation or interruption of breastfeeding during

the study period. However, given that all the drugs in question are used clinically among women who live in regions of the world where breastfeeding is the only affordable, feasible, acceptable, sustainable, or safe option, a better approach is to enroll breastfeeding participants who require the medication for their own health. At the Infectious Diseases Institute (Kampala, Uganda), we have previously enrolled over 350 mother–infant pairs in lactation pharmacokinetic studies evaluating key anti-infective drugs, sampling maternal plasma, maternal breast milk, and infant plasma [61–64]. From our experience, emphasizing and clearly explaining the benefits of characterizing drug exposure of a breastfed infant to the ethics committee during study approval, and while obtaining maternal consent, has enhanced the acceptance of these studies over any presumed ethical obstacles. There is a need to share experiences and competencies between different researchers working on lactation pharmacokinetic studies to streamline the conduct of these studies to define best practices from both a bioethical and pharmacokinetic perspective.

Conclusions

The information on plasma-to-breast milk transfer of drugs commonly used to treat malaria, tuberculosis, and NTDs, and the resultant infant exposure to these drugs through breast milk, is currently limited. The available drug-specific lactation pharmacokinetic data describe maternal blood and breast milk exposures based on data from few mothers and estimate infant drug exposure based on maternal breast milk exposure. Model-based compartmental analyses are increasingly being used to link the dynamic changes in maternal plasma concentrations with breast milk concentrations and characterize interindividual variability in drug concentration-time profiles from fewer samples, compared to non-compartmental analysis. There is an urgent need to build consensus and establish best-practice guidelines for conducting and reporting lactation pharmacokinetic studies to enhance the understanding of plasma-to-breast milk transfer of drugs, determine influential patient- and treatment-related factors, and optimize treatment in mother–infant pairs.

Supporting information

S1 Table. Assessment of clinical pharmacology lactation studies based on the ClinPK evaluation approach.

(DOCX)

S2 Table. PRISMA checklist. (PDF)

Author Contributions

Conceptualization: Francis Williams Ojara, Aida N. Kawuma, Catriona Waitt.

Data curation: Francis Williams Ojara, Aida N. Kawuma, Catriona Waitt.

Formal analysis: Francis Williams Ojara.

Investigation: Francis Williams Ojara, Catriona Waitt.

Project administration: Francis Williams Ojara.

Supervision: Catriona Waitt.

Visualization: Francis Williams Ojara.

Writing - original draft: Francis Williams Ojara, Aida N. Kawuma, Catriona Waitt.

Writing - review & editing: Francis Williams Ojara, Aida N. Kawuma, Catriona Waitt.

References

- World Health Organisation (WHO). World malaria report 2021: an in-depth update on global and regional malaria data and trends. 2021 [cited 2022 Nov 23]. Available from: <u>https://apps.who.int/iris/</u> rest/bitstreams/1398397/retrieve.
- Aweeka FT, German PI. Clinical pharmacology of artemisinin-based combination therapies. Clin Pharmacokinet. 2008; 47(2):91–102. https://doi.org/10.2165/00003088-200847020-00002 PMID: 18193915
- 3. World Health Organisation (WHO). eTB Guidelines [cited 2021 Nov 23]. Available from: https://who.tuberculosis.recmap.org/recommendations.
- Trypanasomiasis, human African (sleeping sickness). 2022 Jan 10 [cited 2022 Nov 23]. Available from: https://www.who.int/news-room/fact-sheets/detail/trypanosomiasis-human-african-(sleeping-sickness).
- Czosnykowska-łukacka M, Królak-Olejnik B, Orczyk-Pawiłowicz M. Breast milk macronutrient components. Nutrients. 2018; 10:1893.
- Verstegen RHJ, Ito S. Drugs in lactation. J Obstet Gynaecol Res. 2019; 45(3):522–531. <u>https://doi.org/10.1111/jog.13899</u> PMID: 30663176
- Ito S. Mother and child: Medication use in pregnancy and lactation. Clin Pharmacol and Ther. 2016; 100 (1):8–11. https://doi.org/10.1002/cpt.383 PMID: 27272612
- Bar-Oz B, Bulkowstein M, Benyamini L, Greenberg R, Soriano I, Zimmerman D, et al. Use of antibiotic and analgesic drugs during lactation. Drug Saf. 2003; 26(13):925–935. <u>https://doi.org/10.2165/</u> 00002018-200326130-00002 PMID: 14583068
- Ilett KF, Kristensen JH. Drug use and breastfeeding. Expert Opin Drug Saf. 2005; 4:745–768. <u>https://doi.org/10.1517/14740338.4.4.745 PMID: 16011452</u>
- Olagunju A, Bolaji O, Amara AB, Waitt AJ. Breast milk pharmacokinetics of efavirenz and breastfed infants' exposure in genetically defined subgroups of mother-infant pairs: an observational study. Clin Infect Dis. 2015; 61:453–463. https://doi.org/10.1093/cid/civ317 PMID: 25882300
- Yeung CHT, Fong S, Malik PRV, Edginton AN. Quantifying breast milk intake by term and preterm infants for input into paediatric physiologically based pharmacokinetic models. Matern Child Nutr. 2020; 16:e12938. https://doi.org/10.1111/mcn.12938 PMID: 31965755
- Nishimura A, Furegen A, Umazume T, Kitamura S, Soma M, Noshiro K, et al. Benzodiazepine concentrations in the breast milk and plasma of nursing mothers: Estimation of relative infant dose. Breastfed Med. 2021; 16(5):424–431. https://doi.org/10.1089/bfm.2020.0259 PMID: 33449825
- 13. Begg EJ, Duffull SB, Hacket LP, Ilet KF. Studying drugs in human milk: Time to unify the approach. J Hum Lact. 2002; 18(4):323–332. https://doi.org/10.1177/089033402237904 PMID: 12449047
- 14. National Library of Medicine. Drugs and Lactation Database 2006 [cited 2022 Nov 25]. Available from: https://www.ncbi.nlm.nih.gov/sites/books/NBK501922/.
- Kanji S, Hayes M, Ling A, Shamseer L, Chant C, Edwards DJ, et al. Reporting guidelines for clinical pharmacokinetic studies: the ClinPK statement. Clin Pharmacokinet. 2015; 54:783–795. <u>https://doi.org/ 10.1007/s40262-015-0236-8 PMID: 25637173</u>
- World Health Guidelines (WHO). Guidelines for the diagnosis and treatment of Chagas disease [cited 2018 Nov 21]. Available from: https://iris.paho.org/bitstream/handle/10665.2/49653/9789275120439_ eng.pdf.
- World Health Guidelines (WHO). Echinococcosis. 2021 May 17 [cited 2022 Nov 26]. Available from: https://www.who.int/news-room/fact-sheets/detail/echinococcosis.
- Pradhan S, Schwartz RA, Patil A, Grabbe S, Goldust M. Treatment options for leishmaniasis. Clin Exp Dermatol. 2022; 47:516–521. https://doi.org/10.1111/ced.14919 PMID: 34480806
- **19.** World Health Guidelines (WHO). Lymphatic filariasis. 2022 Mar 16 [cited 2022 Nov 26]. Available from: https://www.who.int/news-room/fact-sheets/detail/lymphatic-filariasis.
- 20. Welsh O, Al-abdely HM, Salinas-Carmona MC, Fahal AH. Mycetoma medical therapy. PLoS Neg Trop Dis. 2014; 8(10):e3218. https://doi.org/10.1371/journal.pntd.0003218 PMID: 25330342
- Queiroz-telles F, Esterre P, Perez-Blanco M, Vitale RG, Salgado CG, Bonifaz A. Chromoblastomycosis: An Overview of clinical manifestations, diagnosis and treatment. Med Mycol. 2009; 47:3–15. https://doi. org/10.1080/13693780802538001 PMID: 19085206
- 22. World Health Guidelines (WHO). Onchocerciasis. 2022 Jan 11 [cited 2022 Nov 26]. Available from: https://www.who.int/news-room/fact-sheets/detail/onchocerciasis.

- 23. World Health Guidelines (WHO). Scabies. 2022 Aug 16 [cited 2022 Nov 2]. Available from: https://www. who.int/news-room/fact-sheets/detail/scabies.
- 24. World Health Guidelines (WHO). Schistosomiasis. 2022 Jan 8 [cited 2022 Nov 26]. Available from: https://www.who.int/news-room/fact-sheets/detail/schistosomiasis.
- World Health Guidelines (WHO). Taeniasis / Cysticercosis. 2022 Jan 11 [cited 2022 Nov 26]. Available from: https://www.who.int/news-room/fact-sheets/detail/taeniasis-cysticercosis.
- World Health Guidelines (WHO). Yaws. 2022 Jan 10 [cited 2022 Nov 26]. Available from: <u>https://www.who.int/news-room/fact-sheets/detail/yaws.</u>
- García-Bournissen F, Moroni S, Marson ME, Moscatelli G, Mastrantonio G, Bisio M, et al. Limited infant exposure to benznidazole through breast milk during maternal treatment for Chagas disease. Arch Dis Child. 2015; 100:90–94. https://doi.org/10.1136/archdischild-2014-306358 PMID: 25210104
- Moroni S, Marson ME, Moscatelli G, Mastrantonio G, Bisio M, Gonzalez N, et al. Negligible exposure to nifurtimox through breast milk during maternal treatment for Chagas Disease. PLoS Neg Trop Dis. 2019; 13(8):e0007647. https://doi.org/10.1371/journal.pntd.0007647 PMID: 31415566
- 29. Gilder ME, Hanpithakphong W, Hoglund RM, Tarning J, Win HH, Hilda N, et al. Primaquine pharmacokinetics in lactating women and breastfed infant exposures. Clin Infect Dis. 2018; 67(7):1000–1007. <u>https://doi.org/10.1093/cid/ciy235</u> PMID: <u>29590311</u>
- Court R, Gausi K, Mkhize B, Wiesner L, Waitt C, McIlleron H, et al. Bedaquiline exposure in pregnancy and breastfeeding in women with rifampicin-resistant tuberculosis. Br J Clin Pharmacol. 2022; 88:3548–3558. https://doi.org/10.1111/bcp.15380 PMID: 35526837
- Ogunbona FA, Onyyeji CO, Bolaji OO, Torimiro SEA. Excretion of chloroquine and desethylchloroquine in human milk. Br J Clin Pharmacol. 1987; 23:473–476. <u>https://doi.org/10.1111/j.1365-2125.1987</u>. tb03078.x PMID: 3580253
- **32.** Edstein MD, Veenendaal JR, Newman K, Hyslop R. Excretion of chloroquine, dapsone and pyrimethamine in human milk. Br J Clin Pharmacol. 1986; 22: 733–735. https://doi.org/10.1111/j.1365-2125. 1986.tb02967.x PMID: 3567020
- Ogbuokiri JE, Ozumba BC, Okonkwo PO. Ivermectin levels in human breast milk. Eur J Clin Pharmacol. 1993; 45:389–390. https://doi.org/10.1007/BF00265962 PMID: 8299677
- Putter J, Held F. Quantitative studies on the occurence of praziquantel in milk and plasma of lactating women. Eur J Drug Metab Pharmacokinet. 1979; 4(4):193–198.
- Salman S, Davis TME, Page-Sharp M, Camara B, Oluwalana C, Bojang A, et al. Pharmacokinetics of transfer of azithromycin into the breast milk of African mothers. Antimicrob Agents Chemother. 2016; 60 (3):1592–1599.
- Sutton AL, Acosta EP, Larson KB, Kerstner-Wood CD, Tita AT, Biggio JR. Perinatal pharmacokinetics of azithromycin for cesarean prophylaxis. Am J Obstet Gynecol. 2015; 212:812. https://doi.org/10.1016/ j.ajog.2015.01.015 PMID: 25595580
- Abdel-Tawab AM, Bradley M, Ghazaly EA, Horton J, El-Setouhy M. Albendazole and its metabolites in the breast milk of lactating women following a single oral dose of albendazole. Br J Clin Pharmacol. 2009; 68(5):737–742. https://doi.org/10.1111/j.1365-2125.2009.03524.x PMID: 19916998
- Venkatesan K, Mathur A, Girdhar A, Girdhar BK. Excretion of clofazimine in human milk in leprosy patients. Lep Rev. 1997; 68:242–246. https://doi.org/10.5935/0305-7518.19970033 PMID: 9364825
- Edstein MD, Veenendaal JR, Hyslop R. Excretion of Mefloquine in Human Breast Milk. Chemotherapy. 1988; 34:165–169. https://doi.org/10.1159/000238566 PMID: 3262044
- Phillips RE, Looareesuwan S, White NJ, Silamut K, Kietinun S, Warrell DA. Quinine pharmacokinetics and toxicity in pregnant and lactating women with falciparum malaria. Br J Clin Pharmacol. 1986; 21:677–683. https://doi.org/10.1111/j.1365-2125.1986.tb05233.x PMID: 3527243
- Steen B, Rane A. CLINDAMYCIN PASSAGE INTO HUMAN MILK. Br J Clin Pharmacol. 1982; 13:661– 664. https://doi.org/10.1111/j.1365-2125.1982.tb01433.x PMID: 7082533
- Singh N, Golani A, Patel Z, Maitra A. Transfer of isoniazid from circulation to breast milk in lactating women on chronic therapy for tuberculosis. Br J Clin Pharmacol. 2008; 65(3):418–422. <u>https://doi.org/ 10.1111/j.1365-2125.2007.03061.x PMID: 18093257</u>
- Bustinduy AL, Kolamunnage-Dona R, Mirochnick MH, Capparelli EV, Tallo V, Acosta LP. Population pharmacokinetics of praziquantel in pregnant and lactating Filipino women infected with Schistosoma japonicum. Antimicrob Agents Chemother. 2020; 64(9):e00566–20. <u>https://doi.org/10.1128/AAC.</u> 00566-20 PMID: 32631820
- Moore BR, Salman S, Benjamin J, Page-Sharp M, Yadi G, Batty KT, et al. Pharmacokinetics of piperaquine transfer into the breast milk of Melanesian mothers. Antimicrob Agents Chemother. 2015; 59 (7):4272–4278. https://doi.org/10.1128/AAC.00327-15 PMID: 25963980

- 45. Padro JM, Vidal RCP, Echevarria RN, Califano AN, Reta MR. Development of an ionic-liquid-based dispersive liquid–liquid microextraction method for the determination of antichagasic drugs in human breast milk: Optimization by central composite design Chagas. J Sep Sci. 2015; 38:1591–1600.
- Larsen LA, Ito S, Koren G. Prediction of milk / plasma concentration ratio of drugs. The Ann Pharmacother. 2003; 37:1299–1306. https://doi.org/10.1345/aph.1C379 PMID: 12921514
- Pain JB, Le MP, Caseris M, Amiel C, Lassel L, Charpentier C, et al. Pharmacokinetics of dolutegravir in a premature neonate after HIV treatment intensification during pregnancy. Antimicrob Agents Chemother. 2015; 59(6):3660–3662. https://doi.org/10.1128/AAC.00173-15 PMID: 25845873
- Rodari P, Buonfrate D, Pomari E, Lunardi G, Bon I, Bisoffi Z, et al. Ivermectin concentration in breast milk of a woman with Strongyloides stercoralis and human T-lymphotropic virus-I co-infection. Acta Tropica 2020; 202:105249. https://doi.org/10.1016/j.actatropica.2019.105249 PMID: 31678122
- Noe DA. Parameter estimation and reporting in noncompartmental analysis of clinical pharmacokinetic data. Clin Pharmacol Drug Dev. 2020; 9(1):S5–S35. https://doi.org/10.1002/cpdd.810 PMID: 32706165
- Owen JS, Fiedler-Kelly J. Introduction to population pharmacokinetic/pharmacodynamic analysis with Nonlinear Mixed Effects Modelling. 1st ed. Hoboken, New Jersey: John Wiley & Sons, Inc.; 2014.
- Upton RN, Foster DJR, Abuhelwa AY. An Introduction to physiologically-based pharmacokinetic models. Pediatr Anesth. 2016; 26:1036–1046. <u>https://doi.org/10.1111/pan.12995</u> PMID: <u>27550716</u>
- Abduljalil K, Pansari A, Ning J, Jamei M. Prediction of drug concentrations in milk during breastfeeding, integrating predictive algorithms within a physiologically - based pharmacokinetic model. CPT Pharmacometrics Syst Pharmacol. 2021; 10:878–889. https://doi.org/10.1002/psp4.12662 PMID: 34213088
- Ring A, Breithaupt-Grögler K. How publication guidelines for clinical pharmacology trials may help to accelerate knowledge transfer. Br J Clin Pharmacol. 2018; 84:611–614. <u>https://doi.org/10.1111/bcp.</u> 13489 PMID: 29427380
- Moher D., Hopewell S, Schulz K, Montori V, Gotzche PC, Devereaux PJ, et al. CONSORT 2010 Explanation and Elaboration: updated guidelines for reporting parallel group randomised trials. Br Med J. 2010; 340:c869.
- 55. European Medicines Agency (EMA). Guideline on reporting the result of population pharmacokinetic analyses. June 2007 [cited 2022 Nov 26]. Available from: https://www.ema.europa.eu/documents/ scientific-guideline/guideline-reporting-results-population-pharmacokinetic-analyses_en.pdf.
- Dykstra K, Mehrotra N, Tornoe CW, Kastrissios H, Patel B, Al-Huniti N, et al. Reporting guidelines for population pharmacokinetic analyses. J Pharmacokinet Pharmacodyn. 2015; 42:301–314. <u>https://doi.org/10.1007/s10928-015-9417-1</u> PMID: 25925797
- 57. United States Department of Health and Human Services, Food and Drug Administration (USFDA). Clinical Lactation Studies: Considerations for Study Design Guidance for Industry. Draft Guid. May 2019. Available from: https://www.fda.gov/regulatory-information/search-fda-guidance-documents/ clinical-lactation-studies-considerations-study-design.
- Anderson PO. Guidelines for Reporting Cases of Medication Use During Lactation. Breastfeed Med. 2022; 17:93–97. https://doi.org/10.1089/bfm.2021.0357 PMID: 35073165
- 59. Waitt CJ, Garner P, Bonnett LJ, Khoo SH, Else LJ. Is infant exposure to antiretroviral drugs during breastfeeding quantitatively important? A systematic review and meta-analysis of pharmacokinetic studies. J Antimicrob Chemother. 2015; 70:1928–1941. https://doi.org/10.1093/jac/dkv080 PMID: 25858354
- Aebi-popp K, Kahlert CR, Crisinel P, Decosterd L, Saldanha SA, Hoesli I, et al. Transfer of antiretroviral drugs into breastmilk: a prospective study from the Swiss Mother and Child HIV Cohort Study. J Antimicrob Chemother. 2022; 2022(77):3436–3442. https://doi.org/10.1093/jac/dkac337 PMID: 36177836
- Waitt C, Orrell C, Walimbwa S, Singh Y, Kintu K, Simmons B, et al. Safety and pharmacokinetics of dolutegravir in pregnant mothers with HIV infection and their neonates : A randomised trial (DolPHIN-1 study). PLOS Med. 2019; 16(9):e1002895. <u>https://doi.org/10.1371/journal.pmed.1002895</u> PMID: <u>31539371</u>
- Waitt C, Olagunju A, Nakalema S, Kyohaire I, Owen A, Lamorde M, et al. Plasma and breast milk pharmacokinetics of emtricitabine, tenofovir and lamivudine using dried blood and breast milk spots in nursing African mother–infant pairs. J Antimicrob Chemother. 2018; 73:1013–1019. <u>https://doi.org/10.1093/jac/dkx507</u> PMID: 29309634
- Kintu K, Malaba TR, Nakibuja J, Papamicheal C, Colbers A, Byrne K, et al. Dolutegravir versus efavirenz in women starting HIV therapy in late pregnancy (DoIPHIN-2): an open-label, randomised controlled trial. Lancet HIV. 2020; 7:e332–e339. https://doi.org/10.1016/S2352-3018(20)30050-3 PMID: 32386721

64. Pertinez H, Nakalema S, Amara A, Kyohairwe I, Nakijoba R, Lamorde M, et al. Population pharmacokinetic model of lamivudine in 40 Ugandan post-partum mothers. CPT Pharmacometrics Syst Pharmacol. 2020; 9(S1):24. https://doi.org/10.1002/psp4.12497